Author’s response to reviews

Title: ERα-related chromothripsis enhances concordant gene transcription on chromosome 17q11.1-q24.1 in luminal breast cancer

Authors:

Chun-Lin Lin (chunlin.lin2014@gmail.com)
Xi Tan (tanx3@uthscsa.edu)
Meizhen Chen (chenm1@uthscsa.edu)
Meena Kusi (KusiM@livemail.uthscsa.edu)
Chia-Nung Hung (hungc@uthscsa.edu)
Chih-Wei Chou (chouc@uthscsa.edu)
Ya-Ting Hsu (ellenyatinghsu@gmail.com)
Chiou-Miin Wang (wange10@uthscsa.edu)
Nameer Kirma (kirma@uthscsa.edu)
Chun-Liang Chen (chenc4@uthscsa.edu)
Ching-Hung Lin (chinghlin@ntu.edu.tw)
Kate Lathrop (lathrop@uthscsa.edu)
Richard Elledge (elledge@uthscsa.edu)
Virginia Kaklamani (kaklamani@uthscsa.edu)
Kohzoh Mitsuya (mitsuya@uthscsa.edu)
Tim Huang (huangt3@uthscsa.edu)

Version: 4 Date: 28 Apr 2020

Author’s response to reviews:

1. We corrected the direct link of GDC Data Portal for the whole-genome sequencing data.
2. We removed the sentence of "Cancer Methylome System" in the Availability of data and materials section, but referred to the published paper of it in the manuscript.